Author's response to reviews

Title: In vivo effects of Pain Relieving Plaster on closed soft tissue injury in rabbit ears

Authors:

YongZhi Wang (wangyongzhi-1978@hotmail.com)
ChunYu Guo (guochunyu66@sina.com)
HongGang Zhong (greatgun_2202@hotmail.com)
WanNian Zhang (316aw@sina.com)
DeLong Wang (wdljoe@hotmail.com)
Xuan Wang (cp217@hotmail.com)
FuHui Dong (fuhuidong@163.com)

Version: 6 Date: 28 March 2008

Author's response to reviews:

1. The equal contributors are corrected.
2. Figures are cropped as closely as possible to minimize white space around the image.